Response to reviewers’ comments

Reviewer 1#
The abstract needs a little improvement. − It needs to be more focused! ⇒ According to the comment, the abstract was corrected.

Science editor
3. Academic norms and rules: The authors didn’t provide written informed consent. Of treatment.
⇒ “Written informed consent was obtained from the patient before the surgical treatment.” has been described.

4. Supplementary comments were added.

5. (1) “Authors Contributions” section was added.

(2) The pictures and figures in this manuscript were not modified and original from the first.

(3) PRIMD and DOI numbers were added in the all references.

(4) According to instruction of “CASE Presentation” section, “Final diagnosis”, “Treatment”, and “Outcome and follow-up” were added.